Author's response to reviews

Title: Biomechanical Comparison of Unilateral and Bilateral Pedicle Screws Fixation for Transforaminal Lumbar Interbody Fusion after Decompressive Surgery -- A Finite Element Analysis

Authors:

Shih-Hao Chen (shihhao603@tzuchi.com.tw)
Shang-Chih Lin (orthodent.cax@gmail.com)
Wen-Chi Tsai (dtorth05@yahoo.com.tw)
Chih-Wei Wang (jw87121@hotmail.com)
Shih-Heng Chao (ortholabb@gmail.com)

Version: 4 Date: 26 March 2012

Author's response to reviews: see over
Dear Prof Timothy Shipley,

Please find the revised version of the manuscript entitled "Biomechanical Comparison of Unilateral and Bilateral Pedicle Screws Fixation for Transforaminal Lumbar Interbody Fusion after Decompressive Surgery -- A Finite Element Analysis" and associated figures and tables that I would like to submit for consideration of publication as Full Article in *BMC Musculoskeletal Disorders*. The current version has given a point-by-point response to the concerns of the reviewer. The attached files were the revised version of the previous manuscript. The material discussed in this manuscript has neither published nor submitted for publication elsewhere. No benefit of any kind will be received either directly or indirectly by the authors.

Yours sincerely,

Corresponding Author: Shih-Hao Chen, MD
Department of Orthopaedics, Tzu-Chi General Hospital at Taichung, Taiwan
Tel：886-4-36060666 ext. 2765
Fax：886-4-25362316
Email：shihhao603@tzuchi.com.tw